Author's response to reviews

Title: Focussing Both Eyes On Health Outcomes: Revisiting Cataract Surgery

Authors:

Jennifer C Davis (jennifer.davis@ubc.ca)
Heather McNeill (Heather.mcneill1@gmail.com)
Michael Wasdell (MichaelMWasdell@bridgepointhealth.ca)
Susan Chunick (SusanSusan.Chunick@fraserhealth.ca)
Stirling Bryan (stirling.bryan@ubc.ca)

Version: 2 Date: 5 July 2012

Author's response to reviews: see over
July 5, 2012

Dear BMC Geriatrics Editorial Team,

We are pleased to re-submit our manuscript, “Focussing Both Eyes On Health Outcomes: Revisiting Cataract Surgery” for consideration in BMC Geriatrics.

We have been diligent in incorporating the reviewers’ and editorial team comments. These suggestions are addressed in the ‘Response to Reviewers’ document. These revisions are highlighted using the ‘Track Changes’ function in Microsoft Word. We also attached a clean version of the manuscript with ‘Track Changes’ accepted.

We do hope the detailed response to reviewers is helpful in your evaluation of this revised manuscript and we look forward to clarifying further as needed.

All authors have no conflict of interest and financial disclosures to declare.

Sincerely,

Dr. Jennifer Davis, PhD
School of Population and Public Health
Centre for Clinical Epidemiology and Evaluation
VCH Research Institute | The University of British Columbia
Research Pavilion
7th floor, 828 West 10th Avenue
Vancouver, BC V5Z 1M9
Tel: 604-875-4111 ext. 66464 | Fax: 604-875-5179
Jennifer.davis@ubc.ca http://www.c2e2.vchri.ca